Author's response to reviews

Title: A multi-site controlled trial of a cognitive skills programme for mentally disordered offenders

Authors:

Angharad Rees-Jones (angharad.reesjones@gmail.com)
Gisli H Gudjonsson (gisli.gudjonsson@kcl.ac.uk)
Susan J Young (susan.young@kcl.ac.uk)

Version: 2 Date: 28 November 2011

Author's response to reviews:

In the Method section, under the Procedure subheading, we have added a statement providing the name of the ethics committee that approved the research and the reference number.